Author’s response to reviews

Title: Posterior decompression and occipitocervical fixation followed by intraoperative vertebroplasty for metastatic involvement of the axis

Authors:

Xinjie WU (wuxinjie_ms@163.com)
Yingna QI (ayingqi119925@163.com)
mingsheng tan (doctortan_123@126.com)
Ping YI (dryiping_123@163.com)
Feng YANG (dryangfeng@126.com)
Xiangsheng TANG (drtangxiangsheng@126.com)
Qingying HAO (drhaoqingying@126.com)

Version: 5 Date: 29 Dec 2017

Author’s response to reviews:

1. In the ethics approval and consent to participate section of the Declarations, Please include a statement on whether consent to participate in this study was obtained from all patients included in this study. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Authors: This has been added in P11L234-235.

2. In the consent for publication section of the Declarations, please clarify further whether written consent was obtained to publish clinical information and any accompanying images in this paper.

Authors: This has been added in P11L237-238.